Author's response to reviews

Title: Serum paraoxonase and arylesterase activities in patients with lung cancer in a Turkish population

Authors:

Emin T. Elkiran (telkiran@yahoo.com)
Nefsal Mar (nefsalmarh@yahoo.com)
Bilge Aygen (btaygen@yahoo.com)
Ferit Gursu (fegursu@yahoo.com)
Aziz Karaoglu (azizk6@yahoo.com)
Suleyman Koca (kocassk@yahoo.com)

Version: 9  Date: 12 March 2007

Author's response to reviews: see over
Dear Dr Iratxe Puebla,

Title: “Serum paraoxonase and arylesterase activities in patients with lung cancer in a Turkish population”

Dear editor,

Thank you for your letter about resubmitted version of our manuscript.

In response to your suggestion we have further improved the English of the manuscript (highlighted) throughout the text in final version.

In addition, we have also included acknowledgements after the Authors' contributions, as suggested.

We hope we have made satisfactory revisions to our paper to be published in your journal.

Many thanks and best regards

Dr. Emin Tamer ELKIRAN